Author’s response to reviews

Title: The expression of sirtuins, superoxide dismutase, and lipid peroxidation status in peripheral blood from patients with diabetes and hypothyroidism

Authors:

Abdullah Alkhaldi (abmk2004@hotmail.com)

Samar Sultan (sasultan@kau.edu.sa)

Version: 2 Date: 26 Sep 2018

Author’s response to reviews:

1. We note that the current submission contains some textual overlap with other previously published works, in particular:


This overlap mainly exists in the methods, results and conclusion sections of your manuscript.

While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work.

Please rephrase these sections to minimise overlap.

Author response: This now has been re-phrased